Multiparametric MRI to improve detection of prostate cancer compared with transrectal ultrasound-guided prostate biopsy alone: the PROMIS study

Louise Clare Brown, ¹ Hashim U Ahmed, ^{2,3*}
Rita Faria, ⁴ Ahmed El-Shater Bosaily, ^{2,3}
Rhian Gabe, ⁵ Richard S Kaplan, ¹ Mahesh Parmar, ¹
Yolanda Collaco-Moraes, ¹ Katie Ward, ¹
Richard Graham Hindley, ⁶ Alex Freeman, ⁷
Alexander Kirkham, ⁸ Robert Oldroyd, ⁹ Chris Parker, ¹⁰
Simon Bott, ¹¹ Nick Burns-Cox, ¹² Tim Dudderidge, ¹³
Maneesh Ghei, ¹⁴ Alastair Henderson, ¹⁵
Rajendra Persad, ¹⁶ Derek J Rosario, ¹⁷ Iqbal Shergill, ¹⁸
Mathias Winkler, ¹⁹ Marta Soares, ⁴ Eldon Spackman, ⁴
Mark Sculpher ⁴ and Mark Emberton ^{2,3} on behalf of the PROMIS Study Group

¹Medical Research Council Clinical Trials Unit, University College London, London, UK

²Division of Surgery and Interventional Science, Faculty of Medical Sciences, University College London, London, UK

³Department of Urology, University College London Hospitals NHS Foundation Trust, London, UK

⁴Centre for Health Economics, University of York, York, UK

⁵Hull York Medical School and Department of Health Sciences, University of York, York, UK

⁶Department of Urology, Hampshire Hospitals NHS Foundation Trust, Basingstoke, UK

⁷Department of Histopathology, University College London Hospitals NHS Foundation Trust, London, UK

⁸Department of Radiology, University College London Hospitals NHS Foundation Trust, London, UK

⁹Public and patient representative, Nottingham, UK

¹⁰Department of Academic Urology, Royal Marsden Hospital, Sutton, UK

¹¹Frimley Park Hospital, Frimley, UK

¹²Musgrove Park Hospital, Taunton, UK

- ¹³Southampton General Hospital, Southampton, UK
- ¹⁴Department of Urology, Whittington Hospital, London, UK
- ¹⁵Maidstone Hospital, Maidstone, UK
- ¹⁶Bristol Urological Institute, Southmead Hospital, Bristol, UK
- ¹⁷Royal Hallamshire Hospital, Sheffield, UK
- ¹⁸Wrexham Maelor Hospital, Wrexham, UK
- ¹⁹Charing Cross Hospital, Imperial College London, London, UK

Declared competing interests of authors: Hashim U Ahmed reports grants and personal fees from SonaCare Medical, Sophiris Bio Inc. and Galil Medical and grants from TROD Medical outside the submitted work. He conducts private practice in London for the diagnosis and treatment of patients with potential prostate cancer through insurance schemes and self-paying patients in addition to his NHS practice. Ahmed El-Shater Bosaily reports grants and non-financial support from University College London Hospitals/University College London Biomedical Research Centre, grants and non-financial support from The Royal Marsden NHS Foundation Trust and the Institute for Cancer Research Biomedical Research Centre and non-financial support from the Medical Research Council's Clinical Trials Unit during the conduct of the study. Mark Emberton reports grants from Sophiris Bio Inc., grants from TROD Medical, other support from STEBA Biotech, grants and other support from SonaCare Medical and other support from Nuada Medical and London Urology Associates, outside the submitted work. Rita Faria reports grants from the National Institute for Health Research during the conduct of the study. Richard Graham Hindley reports payment as a Clinical Director from Nuada Medical during the period of recruitment. Alexander Kirkham reports other support from Nuada Medical outside the submitted work. Derek J Rosario reports personal fees from Ferring Pharmaceuticals Ltd and grants from Bayer Pharmaceuticals Division, outside the submitted work. Iqbal Shergill reports grants from Ipsen and Astellas Pharma Inc. and non-financial support from Boston Scientific and Olympus, outside the submitted work. Eldon Spackman reports personal fees from Astellas Pharma Canada, Inc., outside the submitted work. Mathias Winkler reports grants and personal fees from Zicom-Biobot (Singapore), outside the submitted work.

Published July 2018 DOI: 10.3310/hta22390

Plain English summary

The PROMIS study

Health Technology Assessment 2018; Vol. 22: No. 39

DOI: 10.3310/hta22390

NIHR Journals Library www.journalslibrary.nihr.ac.uk

^{*}Corresponding author hashim.ahmed@imperial.ac.uk

Plain English summary

The only way to tell whether or not a man has prostate cancer is to examine samples of the prostate under a microscope. These are normally obtained using needles inserted via the rectum, guided by ultrasound imaging. This is called transrectal ultrasound (TRUS)-guided biopsy. However, the ultrasound locates only the prostate and cannot show the cancer, so the needles may miss the cancer.

Transrectal ultrasound-guided biopsy may detect small, slow-growing cancers that would never develop far enough to cause problems. Detecting and treating these has no benefit and exposes men to unnecessary pain and side effects caused by TRUS-guided biopsy. It also creates unnecessary expense for the health service.

This study aimed to find out whether or not a magnetic resonance imaging (MRI) scan can show which men need a TRUS-guided biopsy and which do not. It also assessed whether or not using MRI would improve the detection of cancers that need treatment. Participants volunteered to undergo an extra procedure that takes samples from right across the prostate, as a reference to assess the accuracy of TRUS-guided biopsy and MRI.

We found that using a MRI scan to decide whether or not a biopsy is needed reduces the number of men who have unnecessary biopsies. It may also improve the detection of cancers that need treatment so that fewer are missed. Economic modelling showed that using MRI to decide who should have a TRUS-guided biopsy and to guide the biopsy would be a cost-effective use of NHS resources, providing certain conditions were fulfilled.

HTA/HTA TAR

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.513

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 09/22/67. The contractual start date was in April 2016. The draft report began editorial review in November 2016 and was accepted for publication in July 2017. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2018. This work was produced by Brown et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Director of the NIHR Dissemination Centre, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk